%%
%% This is file `research_contributions.tex',
%% generated with the docstrip utility.
%%
%% The original source files were:
%%
%% sshrc-insight.dtx  (with options: `researchcontributions')
%% 
%% This is a generated file.
%% 
\documentclass[insight_proposal]{subfiles}
\begin{document}
\subfiletitle{Research Contributions}

\subfilesection{1. Relevant research contributions over the last six years}

%% Delete any of the subsections below that do not apply to your
%% research contributions.

\subfilesubsection{Refereed contributions}

\begin{refcontext}[sorting=ydnt]
\begin{refsection}{}
  \nocite{% Add citation keys to your "refereed contributions" here.
  }
\begin{sloppypar}
    \printbibliography[heading=none]
\end{sloppypar}
\end{refsection}{}
\end{refcontext}

\subfilesubsection{Other refereed contributions}

\begin{refcontext}[sorting=ydnt]
\begin{refsection}{}
  \nocite{% Add citation keys to your "other refereed contributions" here.
  }
\begin{sloppypar}
    \printbibliography[heading=none]
\end{sloppypar}
\end{refsection}{}
\end{refcontext}

\subfilesubsection{Nonrefereed contributions}

\begin{refcontext}[sorting=ydnt]
\begin{refsection}{}
  \nocite{% Add citation keys to your "nonrefereed contributions" here.
  }
\begin{sloppypar}
    \printbibliography[heading=none]
\end{sloppypar}
\end{refsection}{}
\end{refcontext}

\subfilesubsection{Forthcoming contributions}

\begin{refcontext}[sorting=ydnt]
\begin{refsection}{}
  \nocite{% Add citation keys to your "forthcoming contributions" here.
  }
\begin{sloppypar}
    \printbibliography[heading=none]
\end{sloppypar}
\end{refsection}{}
\end{refcontext}

\subfilesubsection{Creative outputs}

\begin{refcontext}[sorting=ydnt]
\begin{refsection}{}
  \nocite{% Add citation keys to your "creative outputs" here.
  }
\begin{sloppypar}
    \printbibliography[heading=none]
\end{sloppypar}
\end{refsection}{}
\end{refcontext}

\subfilesection{2. Other research contributions}

%% Enter section text here.

\subfilesection{3. Most significant career research contributions}

%% Enter section text here.

\subfilesection{4. Contributions to training}

%% Enter section text here.

\end{document}
\endinput
%%
%% End of file `research_contributions.tex'.
